Author's response to reviews

Title: Caregivers' active role in palliative home care - to encourage or to dissuade? A qualitative descriptive study

Authors:

    Anna Weibull (daugaardweibull@dadlnet.dk)
    Frede Olesen (fo@alm.au.dk)
    Mette Asbjoern Neergaard (man@alm.au.dk)

Version: 6 Date: 10 September 2008

Author's response to reviews: see over
Concerning MS: 7458005661924524

Caregivers' active role in palliative home care - to encourage or to dissuade?  
A qualitative descriptive study  
Anna Weibull, Frede Olesen, Mette Asbjoern Neergaard

Dear Editor and Editorial Production Team

This time we only made the following four changes:

1. In the Method section, under Sample: Line 6: “The demographics are seen in table 1” are changed to: “The demographics are seen in table 2”

2. In the Method section, under Ethics: Line 9: “and an interview were arranged in the home of the spouse” is changed to: “and an interview was arranged in the home of the spouse”

3. In the Method section, under Ethics: Line 13: “the palliative course of disease has influenced on bereavement” is changed to: “the palliative course of disease had influenced bereavement”

4. In the Discussion section, Line 1: This study indicated” is changed to: “This study indicates”

On behalf of the authors,  
Yours Sincerely

Mette Asbjørn Neergaard  
GP, ph.d.-student  
Phone: +45 8942 6028  
Email: man@alm.au.dk